Author's response to reviews

Title: No role for Estrogen Receptor 1 Gene Intron 1 Pvu II and Exon 4 C325G Polymorphisms in Migraine Susceptibility

Authors:

Natalie J Colson (n.colson@griffith.edu.au)
Rod A Lea (Rod.Lea@vuw.ac.nz)
Sharon Quinlan (s.quinlan@griffith.edu.au)
Lyn R Griffiths (l.griffiths@griffith.edu.au)

Version: 5 Date: 7 February 2006

Author's response to reviews: see over
Dear Members of the Editorial Team

We hereby submit a final manuscript entitled “No role for Estrogen Receptor 1 Gene Intron 1 Pvu II and Exon 4 C325G Polymorphisms in Migraine Susceptibility” by Natalie J Colson, Rod A Lea, Sharon Quinlan, and Lyn R Griffiths.

We have now corrected the formatting, made some minor changes to the acknowledgements and included an Authors' contributions section as outlined below.

Authors' contributions
NC carried out the majority of the genotyping and data analysis and drafted the manuscript. The data analysis was supervised by RL, who also helped draft the manuscript. SQ oversaw recruitment, diagnosis, and clinical characterization of the subjects used in this study. LG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

Acknowledgements
This work was supported by funding from NHMRC and the Australian Brain Foundation. NC was supported by a Griffith University Postgraduate Scholarship and RL by an NHMRC CJ Martin Postdoctoral Fellowship.

We believe that the paper is now ready for publication in BMC Medical Genetics.

We thank you in advance for acceptance of our manuscript.

Yours sincerely

Professor Lyn Griffiths
Director, Genomics Research Centre